Author's response to reviews

Title: Transplantation of Endothelial Progenitor Cells in Treating Rats with IgA Nephropathy

Authors:

Wei Guo (testement@163.com)
Jiang-min Feng (fengjiangm@163.com)
Li Yao (liyao_cmu@163.com)
Li Sun (sunlicmu1974@163.com)
Guang-qin Zhu (zhuguangqinys@sina.com)

Version: 2
Date: 4 November 2013

Author's response to reviews: see over
November, 2013

Dear Editor:

Our submitted manuscript entitled “Transplantation of Endothelial Progenitor Cells in Treating Rats with IgA Nephrology” by Wei Guo, Jiang-Min Feng, Li Yao, Li Sun, Guang-Qing Zhu, to be considered for publication in the BMC Nephrology

According to Editorial request, we made some changes to the formatting of our manuscript.

1. Add “Conflict of interest statement” before Acknowledgements
2. Add “Authors' Contribution” before Acknowledgements
3. Give the emails of all the authors in the title page
4. Add conclusion section after discussion

Thank you very much for your consideration.

Sincerely,

Wei Guo and Jiang-Min Feng (corresponding author)
And affiliation: Department of Nephrology, The First Affiliated Hospital of China Medical University, Shenyang110001, China